Author's response to reviews

Title: Sun protection for children with oculocutaneous albinism in South Africa; health policy issues

Authors:

Patricia M Lund (p.lund@coventry.ac.uk)
Julie S Taylor (j.z.taylor@dundee.ac.uk)

Version: 2 Date: 29 November 2007

Author's response to reviews:

Details of the permissions granted are now detailed in the methods section. The study was undertaken in a rural area of Southern Africa where ethical approval was granted in accordance with the country's governance procedures - in this case the higher education authority.